Author's response to reviews

Title: Feasibility and metabolic effects of carbohydrate loading in patients with fragility hip fracture: study protocol for a randomised controlled trial Pre-Operative nutrition In Neck of femur Trial (POINT).

Authors:

Iain K Moppett (iain.moppett@nottingham.ac.uk)
Paul Greenhaff (paul.greenhaff@nottingham.ac.uk)
Ben J Ollivere (ben.ollivere@nuh.nhs.uk)
Theophillus Joachim (theojoachim28@gmail.com)
Dileep N Lobo (dileep.lobo@nottingham.ac.uk)
Martin Martin Rowlands (mrowlands1@live.co.uk)

Version: 2 Date: 18 July 2014

Author's response to reviews:

Changes made as requested by editorial team:

1. Please include the date of registration with your trial registration number at the end of the Abstract.

2. Please include a statement in your Methods section explaining that you obtained informed consent from each participant.

3. Please move your statement of ethical approval to the Methods section.

4. If appropriate, please include a list of abbreviations used and their meanings, after your Trial Status.